Author’s response to reviews

Title: Magnitude and determinants for place of postnatal care utilization among mothers who delivered at home in Ethiopia: A multinomial analysis from the 2016 Ethiopian Demographic Health Survey.

Authors:

Brhane Ayele (brhane3127@gmail.com)
Mulugeta Woldu (mulugetawoldu425@gmail.com)
Haftom Gebrehiwot (haftom1224@gmail.com)
Equbay Gebre-egziabher (equblab2003@gmail.com)
Hailay Gebretnsae (hailish14@gmail.com)
Tsegay Hadgu (tsegayhadgu28@gmail.com)
Alemnesh Abrha (alemneshab2017@gmail.com)
Araya Medhanyie (araya.medhanyie@gmail.com)

Version: 2 Date: 30 Aug 2019

Author’s response to reviews:

Dear Dr. Vogel, dear reviewers:

Thank you very much for your attentive follow-up and understanding our previous response. As much as possible here is the revised version of our manuscript after edition is made.

Sincerely yours,

Brhane Ayele (on behalf of the authors)